

This is a repository copy of Interpreting an escape from an eviction trap as a social account: a Gramscian reading of a credit union's policies in support of social housing tenants.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/197686/</u>

Version: Published Version

Article:

Lee, B. orcid.org/0000-0003-2656-4106 and Carlisle, L. (2023) Interpreting an escape from an eviction trap as a social account: a Gramscian reading of a credit union's policies in support of social housing tenants. Critical Perspectives on Accounting. 102582. ISSN 1045-2354

https://doi.org/10.1016/j.cpa.2023.102582

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



Critical Perspectives on Accounting xxx (xxxx) xxx



Contents lists available at ScienceDirect

Critical Perspectives on Accounting



journal homepage: www.elsevier.com/locate/cpa

Interpreting an escape from an eviction trap as a social account: A Gramscian reading of a credit union's policies in support of social housing tenants

Bill Lee^{*}, Liam Carlisle

Accounting and Financial Management Division, Sheffield University Management School, Conduit Road, Sheffield S10 1FL, UK

ARTICLE INFO	A B S T R A C T
Keywords: Critical theory Social accounts Credit unions Social housing London	In capitalist economies, residences that have value as homes are also commodities. Use of Gramsci's concept of historic bloc highlights how in the current, neoliberal period, governments' increasing perception of houses as commodities has affected allocation of social housing and contrasts with the earlier social democratic period when social housing's use as homes was a more prominent consideration. Policy changes in the neoliberal period reduced social housing stock, increased rents and the precarity of income of many people dependent on social housing, particularly in London. Such policies created a trap of eviction if tenants accrued rent arrears. Empirical research reports one credit union's initiatives to ameliorate the threat of tenants'

evaluate the credit union's initiatives.

eviction. Marxist interpretations of social accounts are used to understand the eviction trap and

1. Introduction

Homes that provide safe havens for people, have a dualistic quality in capitalist economies where they are also commodities (Marx, 1954; Wilde, 2020). In recent years, increasing commodification of housing has created "a global "epidemic" of evictions [that] has been unfolding across many cities of the Global South and Global North" (Zamfirescu & Chelcea, 2020, p. 1). The poorest members of society are most likely to be evicted from their homes (Lees & White, 2020; Wilde, 2020; Zamfirescu & Chelcea, 2020). This is evident in London where displacement of the least affluent has led Lees and White (2020) to argue that there has been a social cleansing of the city. Commodification of property may be omnipresent in capitalist economies. However, the precariousness of residents' occupancy of homes varies according to the regulatory protection afforded to tenants (e.g., Watt, 2021) and the extent to which providers of housing are driven by financial criteria over occupiers' rights (for example, Smyth, Cole, & Fields, 2020). The purpose of this article is to explain the need for – and the content, merits, limitations and accounts of – a credit union's policies directed towards preventing eviction of social housing tenants in two London boroughs following governmental policy changes affecting such tenants' housing tenure.

The article analyses two processes to realize the above objective. Firstly, it considers the increased potential for eviction following changes in policies affecting how governments accounted for social housing across two periods in Britain. The first was the "social-democratic settlement" period (Wilde, 2020, p. 4) between the 1940s-1970s when governments accounted for social housing as homes. The second is the neoliberal period since the late nineteen-seventies when governments gave increasing prominence to residences as

* Corresponding author. *E-mail addresses:* w.j.lee@sheffield.ac.uk (B. Lee), advice@pluscu.co.uk (L. Carlisle).

https://doi.org/10.1016/j.cpa.2023.102582

Received 31 March 2022; Received in revised form 7 February 2023; Accepted 21 February 2023

1045-2354/© 2023 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

commodities when accounting for social housing, raising the threat of eviction for some. The second process of interest is the introduction, success, limitations and accounting of policies pursued by one credit union to help social housing tenants threatened by eviction. When analysing those two processes, the article makes four contributions. Firstly, it introduces Gramsci's (1978) idea of a historic bloc into the accounting literature to explain changes in Governments' approaches to accounting for public services, including social housing, between the social democratic period and the neoliberal period. Secondly, it synthesises Marxian arguments about social accounts to understand the detrimental impacts of governments increasingly accounting for public services as commodities and the challenges to countering those impacts. Thirdly, it theorizes an "eviction trap" for some tenants arising from changes to accounting for social housing. Finally, it presents an empirical study of a credit union's initiatives around housing that sought to protect social housing tenants from eviction.

The article makes the following argument. Capitalist economies are marked by periods of hegemony, whereby those who control capital enter an alliance with some subordinate groups to form what Gramsci (1978) describes as a historic bloc across the State, economy and civil society. The historic bloc seeks to articulate conditions for development of the economy under direction of the ruling class. During the social-democratic settlement period, the historic bloc included governments accounting for public services according to need in recognition of ordinary working people's sacrifices made in World Wars. However, hostility to organized labour in some quarters accompanied an economic crisis in the nineteen-seventies and led to formation of a new historic bloc and election of governments that pursued neoliberal principles. Accounting for provision of many services shifted increasingly to users' ability to pay for those services. Application of such policies to social housing provision contributed to a potential trap of eviction for the most vulnerable tenants who experienced increasing rent obligations with limited incomes. Credit unions are financial institutions that do not focus primarily on housing needs, but one credit union took initiatives to militate against the trap of some vulnerable people losing their homes. These initiatives helped protect some from eviction. A fundamental shift back to accounting for social housing as homes is required to overcome the eviction trap. Marxian-informed social accounts may contribute to formation of an alternative historic bloc to achieve such a development.

The remainder of the discussion is as follows. Section 2 outlines the theoretical framework. It uses Gramsci's idea of historic blocs to explain the movement from the social democratic period that accounted for public services according to people's needs, to the neoliberal period that increasingly accounted for public services according to their monetary value and it outlines this article's use of Marxian-informed ideas about social accounting to illuminate those changes and their consequences. Section 3 describes how movement between historic blocs included some shift in focus of accounting for social housing from homes to commodities. This shift – manifest in policies affecting levels of social housing stocks, rents and tenants' incomes – contributed to an eviction trap which was experienced most accutely in London. Section 4 reports empirical research into a London-based credit union's initiatives that sought to protect tenants from the eviction trap. Section 5 concludes by reflecting on how this article's ideas could contribute to the formation of an alternative historic bloc.

2. Theoretical framework - Gramscian thought, historic blocs and social accounting

Gramsci's analysis is prominent in the critical accounting literature (Alawattage & Wickramasinghe, 2008; Cooper, 1995; Haines-Doran, 2022; Goddard, 2002, 2005; Lee, 2010; Lee and Cassell, 2008, 2017; Lehman, 1995; Lehman & Tinker, 1987; Richardson, 1989; Spence, 2009; Smyth & Whitfield, 2017). Gramsci's concept of historic blocs – along with related concepts of hegemony, civil society and organic crisis – can help to trace long-term changes in the British State's policies. As the concept of historic bloc is largely absent from the accounting literature, it is explained briefly here. Gramsci was a Marxist. He saw a dominant capitalist class – in an alliance with some subordinate groups – articulating conditions for development of an economy. This alliance of forces that spanned the economy, politics and civil society constituted a historic bloc. The historic bloc provides the apparatus through which a dominant class exercises hegemony. Hegemony refers to a combination of the dominant class gaining consent from the majority of a population for ideas about the economy's development, with the threat of State coercion against any deviant minority who obstruct those ideas' implementation (Lehman, 1995, pp. 74, 76). Hegemony is realized partly through civil society institutions.

Civil society may be a contested concept (Smyth, 2012a, p. 234), but this discussion accepts Gramsci's (1978, p. 12) definition of "the ensemble of organisms commonly called "private"". Private institutions are those not controlled directly by the dominant class or the State. Examples include families, churches, trade unions, clubs, professional associations and social movements (Burawoy, 2000; Simon, 1982). Civil society institutions provide spaces where ideas about – and consequences of – the organization of an economy are experienced, interpreted and either supported, accepted, challenged, revised, or rejected. These spaces are a contested terrain that may reinvigorate consent for the existing historic bloc, or they may accommodate "long ideological and political preparation" to form a new historic bloc for change (Gramsci, 1978, p. 110; see also, Cohen, 1983, pp. 23 et seq.; Lee & Cassell, 2008; Leversha, 1977, p. 119; Simon, 1982, p. 74; c.f. Cohen & Arato, 1992, p. 151; c.f. Smyth, 2012a, p. 234). Consistent with Marx's (1959) thought, Gramsci saw changes to the organic composition of capital – when manifest in displacement of the variable capital of labour that *creates* new additional value by constant capital involving machinery, buildings, materials, etc. that only *transfer* part of their value to a new product or service – leading to a tendency for the rate of profit to fall and periodic crisis. During the crisis, capital equipment's value reduces markedly, creating the potential for the rate of profit to increase substantially to facilitate advent of a new period of capitalist accumulation (see also, Bryer, 2017, pp 144-5; Kliman, 2007, 2012, pp 14 et seq.; 2015, pp. 241 et seq.; Kliman, 2017; Roberts, 2016, pp 12 et seq.). Gramsci alludes to this as an organic crisis, which necessitates restructuring the economy on more-or-less favourable terms to either capital or labour around a new historic bloc of political alliances.

This framework informs the following analysis of the British economy. The historic bloc that structured the social democratic consensus settlement included representatives and concerns of organized labour manifest in the reforms enacted by the first majority

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

Labour Government in the aftermath of the 1939–45 World War. Workers returning from the 1939–45 conflict were dissatisfied with the poverty and austerity of the nineteen-thirties (see Roberts, 2016, chapter 3 for analysis of the 1930 s Great Depression). The new Labour Government assimilated those workers' political strength and demands into the historic bloc in ways expressive of a greater equality between the classes. Thus, Keynesian demand management policies were used to pursue full employment and tripartite bodies of representatives of labour, owners of capital and government were established to promote industrial consensus. Workers' political strength was also manifest in the State accounting for many services according to need, so there were extensions to free education, introduction of a National Health Service to provide free healthcare when required, provision of benefits to protect people unable to work and expansion of social housing provision. Such reforms generated consent for a State that left the economy under capitalist direction (Simon, 1982, p. 53). All main parties of government supported this settlement that lasted until an organic crisis in the nineteen-seventies when the neoliberal period emerged.

Kliman (2012, pp 50-1) suggests neoliberalism is a useful analytic category for understanding "the dominant politics and ideology of a particular period" of capitalism. Hall (1987) reports how ideas hostile to the social-democratic period were propagated by small right-wing think tanks over time until they gained broader popularity including in the Conservative Party at the onset of an organic crisis. The crisis arose when investment in new technology led to changes in the organic composition of capital that reduced the rate of profit (Roberts, 2016, p. 60). However, right-wing think tanks blamed the crisis on: The State being too bureaucratic to deliver services efficiently; demand management policies causing inflation resulting in British products being uncompetitive internationally; overly powerful trade (i.e., labor) unions forcing wages above their natural market level while preventing industrial innovations; and universal benefit entitlements encouraging a dependency culture. The Conservative government of 1979 began construction of a new historic bloc that excluded labour by abolition of tripartite industrial bodies. Legislative reforms limited trade unions' capacity to organize while economic demand management policies used previously to realise full employment were abandoned which sapped workers' strength. Kliman (2012, 2015; see also, Roberts, 2016) argues destruction of capital did not accompany advent of the neoliberal period, unlike in previous crises, so there has been limited potential for the rate of profit to rise to motivate capitalist expansion. Instead, there has been a long period of stagnation, apart from the occasional counteracting tendencies such as increasing profitability from globalisation in the nineteen-nineties (Carchedi & Roberts, 2013). Nevertheless, successive governments articulated the terms of a new hegemony around neoliberalism which Harvey (2005) explains as assuming markets - and not the State - are appropriate mechanisms for distributing all resources. Consequently, the State has introduced internal markets into many public provisions, made charges for some health services and accounted for others including social housing - which will be discussed in section 3 - according to their monetary values (see also, Ejiogu, Ambituuni, & Ejiogu, 2021).

Professional associations – including those of accountants – are a feature of civil society and may help define and perpetuate a historic bloc. Hitherto, accountants' professional associations helped to develop tools to perpetuate capitalism through successive historic blocs (Goddard, 2002; Lehman & Tinker, 1987). Thus, accounting promoted pursuit of profit for those who control capital and represented workers simply as a cost of production to be reduced (Sikka, Wearing, & Nayak, 1999). Consequently, when financial accounting measurements are applied to institutions organized along other principles, they can subvert more noble goals (for example, Cooper & Johnston, 2012; Gray, Bebbington, & Collison, 2006; Lapsley, 2008). In response, Gray (2002; Gray, Brennan, & Malpas, 2014) proposed social accounts to provide alternative forms of measurement. This discussion divides different theories about the potential for social accounts according to whether they have the explicit objective of transcending the present capitalist social relations of production and their associated forms of exploitation. Some authors (Carroll & Shabana, 2010; Elkington, 1997) suggest aligning financial benefits with social goals, such as by reducing costs of employee turnover by making work roles more interesting and reducing outlay on energy by green initiatives. Such perspectives accept the current social relations of production. Others recognize competing interests and political power (for examples, Bebbington, Brown, Frame, & Thomson, 2007; Brown, 2009, pp. 316–317; Dey, Russell, & Thomson, 2011; Gray, Owen, & Adams, 1996; Thomson, Dey, & Russell, 2015; cf., Catchpowle & Smyth, 2016; cf., Lehman, 1999). They suggest strategies that include (i) pressurizing organizations to prepare accounts of their social and environmental impact for scrutiny and challenge, (ii) preparing Silent or Shadow Accounts about such impacts from information in the public domain to create pressure for change and (iii) establishing dialogical ways of realizing polyvocal expressions of different interests and alternative views of an organization. These perspectives accept endurance of capitalist enterprises and so do not provide a full challenge to the current social relations of production.

Bryer (2017, 2019), Cooper, Taylor, Smith, and Catchpowle (2005) and Lee and Cassell (2008) provide complementary perspectives that suggest a means to supersede capitalist social relations of production. Lee and Cassell highlight how social accounts may help develop civil society institutions to be prefigurative of alternative social relations of production. Cooper et al. propose a political strategy for developing social accounts to promote movement away from a capitalist economy. Cooper et al. argue that violations of social and environmental life are intertwined with positions of power in capitalist economies, so it is necessary for social accounts to be prepared away from influence by markets and management in capitalist economies and informed by critical theory to disrupt the prevailing ideological understandings. Cooper et al. suggest that such social accounts are most effective when aligned with the struggles of social movements to add force to change. Bryer's (2017, 2019) work provides insights into the critical theory to use to inform social accounts. Bryer (2017, p. 45) articulates social accounts in two ways derivative of Marx's work. The first is his employment of Marx's labour value categories to recognise a product or service's use value. Bryer (2019) also argues that each mode of production – such as slavery, feudalism and capitalism – has its own calculative mentality that informs how the dominant class extracts surplus labour from subordinate classes. In capitalist economies, products and services are sold for their exchange values. Those who control capital employ a monetary-based calculative mentality and classifications – or what Bryer describes as phenomenal forms – of "wages", "rent", "costs", etc., to extract surplus labour in the form of "profit". The phenomenal forms overlay the invisible essence of labour value categories and are a social construction largely specific to the capitalist epoch. The socially constructed phenomenal

B. Lee and L. Carlisle

forms provide the second type of social account articulated by Bryer. Consistent with Marx, Bryer argues that capitalism has developed productive capacity exponentially, sufficient to satisfy human needs. However, the use of phenomenal monetary forms in accounting by those who control capital to decide production according to calculations of profitability serve as a fetter on production of use values. Bryer suggests social accounts may inform progressive change by highlighting how using phenomenal forms to account for products and services by their exchange value as commodities, provide fetters on production of use values.

This view of social accounts as highlighting use values and constraints on their production by accounting for exchange values and the potential to align those social accounts with struggles of social movements to inform composition of an alternative historic bloc informs the remainder of the discussion.

3. Context - Changes in historic blocs, accounting for social housing and the eviction trap

Eviction is the unwilling removal of people from their accommodation (Baker, 2020; c.f., OECD, 2021). The term "trap" is employed because eviction takes place due to changes in policies beyond the affected individuals' control. When comparing historic blocs, there is no intention to idealise the social democratic period which promised universal entitlement to use value of social housing, nor to suggest that the neoliberal period has not included some positive initiatives. Some housing in the earlier period was built rapidly in a substandard way (Garrett, Nicol, & Mackay, 2019, p. 9). There was also a more limited eviction trap as the film *Cathy Come Home* highlighted (Crisis, 2018, p. 31). In the latter period, some initiatives in Britain's constituent nations aimed to help some groups threatened with homelessness (Crisis, 2018, chapter 2). However, the overall picture is one of rising numbers of evictions because of

 Table 1

 Number and percentage of new home build starts annually by housing sector.

	Local Council		Other Social Housing		Private		Totals	
Year	Number	%	Number	%	Number	%	Number	%*
1978	75,160	33.2	17,960	7.9	133,580	58.9	226,700	100
1979	55,190	29	14,240	7.5	121,130	63.6	190,560	100.1
1980	33,550	25.9	12,910	9.9	83,300	63.2	129,760	100
1981	21,670	16.5	9,870	7.5	99,610	76	131,150	100
1982	28,870	17.4	14,320	8.6	122,463	74	165,653	100
1983	29,420	15.6	11,760	6.2	147,840	78.2	189,020	100
1984	24,120	14.3	11,220	6.6	133,410	79.1	168,750	100
1985	18,650	11	10,170	6	141,110	83	169,930	100
1986	16,840	9.3	10,910	6	153,990	84.7	181,740	100
1987	16,020	8.1	9,580	4.9	171,200	87	196,800	100
1988	13,440	6.2	10,370	4.8	193,480	89	217,290	100
1989	12,760	7.7	11,000	6.7	141,460	85.6	165,220	100
1990	6,640	5	14,100	10.6	112,730	84.5	133,470	100
1991	3,060	2.3	16,440	12.3	114,310	85.4	133,810	100
1992	1,610	1.2	28,110	21.7	99,590	77	129,310	99.9
1993	1,200	0.8	33,570	22.2	116,460	77	151,230	100
1994	450	0.3	33,590	20.3	131,400	79.4	165,440	100
1995	580	0.4	25,240	18.5	110,410	81	136,230	99.9
1996	490	0.3	22,630	15.6	121,530	84	144,650	99.9
1997	310	0.2	21,190	13.4	136,080	86.4	157,580	100
1998	100	0.1	17,490	11.7	131,810	88.2	149,400	100
1999	170	0.1	17,930	12.1	130,290	87.8	148,390	100
2000	100	0.1	14,040	9.8	128,470	90.1	142,610	100
2001	180	0.1	13,280	9	133,320	90.8	146,780	99.9
2002	160	0.1	14,560	9.7	135,970	90.2	150,690	100
2003	300	0.2	15,600	9.7	145,390	90.1	161,290	100
2004	170	0.1	19,370	11	157,150	89	176,690	100.1
2005	180	0.1	20,920	12	152,800	87.9	173,900	100
2006	290	0.2	21,110	12.4	149,210	87.5	170,610	100.1
2007	150	0.1	23,540	12.8	159,900	87.1	183,590	100
2008	370	0.3	24,160	22.6	82,370	77.1	106,900	100
2009	150	0.2	20,460	23.9	65,000	75.9	85,610	100
2010	1,460	1.3	24,350	22	84,860	76.7	110,670	100
2011	1,700	1.5	23,810	21	87,790	77.5	113,300	100
2012	1,510	1.5	19,260	19.1	80,260	79.4	101,030	100
2013	1,080	0.9	24,900	20	98,820	79.2	124,800	100.1
2014	2,630	1.9	26,340	18.7	111,790	79.4	140,760	100
2015	1,680	1.1	25,900	17.5	120,590	81.4	148,170	100
2016	1,660	1.1	25,340	16.3	128,160	82.6	155,160	100
2017	1,820	1.1	27,000	16.5	135,300	82.4	164,120	100
2018	2,420	1.4	28,010	16.6	138,190	82	168,620	100
2019	1,780	1.2	28,160	18.4	123,060	80.4	153,000	100
2020	1,740	1.3	25,280	19.5	102,400	79.1	129,420	99.9
2021	2,410	1.4	31,270	17.8	141,710	80.8	175,390	100

Source. Gov.UK (2022). * Annual percentages may not add due to rounding.

B. Lee and L. Carlisle

Governments increasing accounting for social housing by its exchange value.

At the advent of the first historic bloc, the new Labour Government saw housing as an integral part of the Welfare State (Watt, 2021). It adopted a policy commitment of "Homes for All" who could not afford the exchange value of houses demanded by commercial builders. The Labour Party's (1945) General Election manifesto promised to "proceed with a housing programme with the maximum practical speed until every family in this island has a good standard of accommodation" and pledged to build four million homes within ten years (Murphy, 1970). The new Labour government's accounting for social housing according to use value for tenants was manifest in its requirement for local government authorities (hereafter councils) to draw up plans for the residences needed by their local populations, provision of funds to assist building and ongoing maintenance of such residences, imposition of rent controls, prohibition of council house sales and – while the communal stock was established – prioritising licenses for council house construction over private sector provision (Murphy, 1970, pp. 415-6). The *1948 National Assistance Act* also required local councils to protect people in need. Between 1946 and 1960, many council houses were constructed (Garrett et al., 2019; Watt, 2021). The *Housing (Homeless Persons) Act 1977* completed the commitment of providing council houses for those who could not afford to purchase homes from commercial builders by requiring councils to offer housing to the homeless, although even at this point some groups were excluded as the legislation made no provision for single people and childless families (Crisis, 2018; Wilde, 2020). Nevertheless, the percentage of public sector homes of Britain's housing stock increased from 10% in 1939 to 30% in 1979 when 42% of the population lived in council houses (Daly & Lund, 2020; Harris, 2016; UK Housing Review, 2021).

The switch to neoliberalism and governments' increasing emphasis on accounting for social housing by exchange values started following the election of the 1979 Conservative Government. Although the ban on council house sales was lifted in 1952 (Murphy, 1970, p 424), the Conservative Government's 1980 Housing Act was a watershed intervention that allowed tenants to buy their council properties at heavily discounted prices (Lees & White, 2020; Wilde, 2020). Almost 2 million council homes have been sold in England since the 1980 Housing Act (Lees & White, 2020). Use value of these houses was removed from a common treasury as councils did not receive sufficient funds for replacement housing. The 1986 Housing and Planning Act allowed councils to transfer their remaining housing to others including registered social landlords (Smyth, 2012b). Governments have also reduced the financial sums provided to councils and others to build new social housing (Boyne & Walker, 1999; Daly & Lund, 2020; Laffin, 2019; Smyth et al., 2020). The impact of these changes on the number of council houses constructed is shown in Table 1 above. As the table illustrates, retreat from public sector provision of around one-third of homes annually started with the neoliberal period, long before the Housing and Planning Act 2016 extended neoliberal emphasis on exchange values by articulating a central principle that the market should satisfy housing needs through either owner-occupation, or private rental accommodation (Lees & White, 2020). Expanded provision of other social housing has not offset the loss of new homes by councils. The table also shows that contrary to the aspirations of the Housing and Planning Act 2016, reduced council house provision is not offset by houses built by commercial builders. Notably, the National Housing Federation (NHF, 2020) estimates 1.6 million households are waiting for social housing provision, despite its declining status since the social democratic period (Ejiogu & Denedo, 2021). Reflecting Bryer's (2019) analysis, this disjunction between need and provision highlights how capitalist relations of production have placed increasing fetters on construction of houses for their use values during the neoliberal period.

Problems associated with growing scarcity by governments increasingly accounting for social housing by exchange value have been accentuated by similar policies manifest in the determination of rents, to help create the eviction trap. In contrast with rent controls to prioritise social housing's use values during the social democratic consensus period, the *1980 Housing Act* introduced a new formula for the Department of Environment to increase tenants' rents, which then rose steeply (Fée, 2009). The *1980 Housing Act* proved to be a harbinger of a longer-term intention for social housing rents to reflect market rates (Daley and Lund, 2020) Upwards pressures on social housing rents may have been increased in recent years by the neoliberal development of "financialization" which is used here simply as "a descriptive term that refers to the relative grown and increased importance of financial markets, instruments and institutions" (Kliman & Williams, 2015, p. 89). Smyth et al. (2020) report how, following the 2007–8 Global Financial Crisis, continuing reductions in central government support led large housing associations to issue bonds to finance provision of social housing. This increases market pressure on social housing providers to prioritise rental exchange value rather than use value for tenants.

Problems of scarcity of housing and increased rents have been compounded by neoliberal policies affecting incomes that have made social tenants' rents increasingly unaffordable. When council house rents were increased initially, sufficient State benefits were promised to help those on low incomes to pay rent. However, subsequent welfare reform in the guise of the *2012 Welfare Reform Act*, which replaced different benefits by a universal credit, was followed by long delays before payments were made (Crisis, 2018). Many tenants fell into rent arrears. Subsequent initiatives further reduced existing tenants' capacity to afford social housing's rents. These changes included reducing rent refunds to a maximum of 80% of the local market rate and introduction of penalties for any unoccupied bedrooms (Lees & White, 2020; Wilde, 2020). One calculation suggests that after receipt of benefits, "between 2009 and 2010 and 2015–2016 council-house rents increased by 37.7 per cent and housing association rents by 38.6 per cent." (Daly & Lund, 2020, p. 420). Freezing of benefits by the *Welfare Reform and Work Act 2016* accentuated the problems (Lees & White, 2020). Clark, Hamilton, Jones, and Muir (2017) put the cumulative cost of those reforms at between £22 and £70 per month on the budgets of poor households outside London. Social housing tenants now spend 50% more of their income on housing compared to those living in their own properties (NAO, 2021). Consequently, there has been a steady increase in the rates of eviction from social housing, primarily because of rent arrears (Lees and While, 2020).

Impacts of governments moving from accounting for social housing as a common treasury of use value to accounting for its exchange value – with the concomitant depletion of housing stock, higher rents and reduced incomes – are felt most acutely in London. During the social-democratic period, several hundred thousand new social housing residences were built. By 1981, 34.8% - or over 870,000 – of homes in London were social houses of which around 770,000 were owned by councils (Lees & White, 2020; Watt, 2021;

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

Wilde, 2020). Around 300,000 council homes in London were sold under tenants' right to buy arrangements (Lees & White, 2020; Wilde, 2020) and the number of rented properties offered by local councils dropped to 392,800 by 2018 (Watt, 2021). Changes to rent policy made social housing increasingly unaffordable when "affordable rents" are defined as up to 80 percent of the local market rate (Department for Community and Local Government, 2014, p. 4) in a city that has some of the highest international land values (Lees & White, 2020). As Wilde (2020, p. 2) comments "London's status as the unrivalled king of the global property league ... drives up prices across the board, leaving those on low incomes in the impossible position of trying to meet rental prices that have been grotesquely inflated by financial speculation". Such property price premiums co-exist with the Regulator of Social Housing (2021) standard requiring housing associations to ensure "optimal benefit is derived from resources and assets", which increases their intolerance of arrears. Reduced entitlements to housing-related benefits are also most stark in central London where poor households experienced reductions in support of between £124 and £1,036 per month (Clark et al., 2017). Consequently, those unable to work or in low paid employment receive insufficient income "to cover rent, council tax, utility bills, food and travel, even when living in low cost council housing" (Lees & White, 2020, p. 1712). Thus, prior to the COVID19 pandemic, evictions in London rose consistently to new record levels, year on year (Wilde, 2020) and 62% of all evictions now take place in London (Clark et al., 2017).

Having outlined how a shift from accounting for social housing according to its use value, to accounting for it according to its exchange value, has contributed to an eviction trap for social housing tenants, the next section considers one credit union's initiatives to help some tenants in London boroughs avoid that trap.

4. The credit union and its housing initiatives

This case study analyses the content, merits and accounts of homeless prevention policies of a credit union – *Selond* (pseudonym) – that supported tenants retaining the use value of their social housing in two London boroughs when caught in the eviction trap. A credit union is a "financial cooperative, democratically controlled by its members, and operated for the purpose of maximizing the economic benefit of its members by providing financial services at competitive and fair rates" (WOCCU, undated). Credit unions may be considered as hybrid organizations. On the one hand, they are community-based, civil society institutions that developed as a social movement of membership cooperatives to counter financial exclusion (Jones, 2005). On the other hand, credit unions function as financial institutions. For example, the *2000 Financial Services and Markets Act* (HMG, 2001) brought credit unions in Britain into a common regulatory regime with commercial banks that assist extraction of surplus value for owners of capital by managing monetary exchanges to generate profits for separate groups of shareholders. This hybridity influences the range of policies reported below whereby *Selond* sought to help protect the use value of tenants' homes within the regulatory and financial constraints of enduring as a viable financial institution in a capitalist economy.

Consistent with other accounting researchers (e.g., Modell, 2009, 2017; Smyth, 2012a), Bhaskar's (1975) critical realism provides the underlying philosophical assumptions for this research. Bhaskar distinguished between real, actual and empirical domains of reality. The real domain refers to underlying structures and processes which may not be observable, but which have consequences manifest in the actual domain. A relevant example of the real domain is Marx's (1959) idea that when labour is recognised as the sole creator of additional value, changes to the organic composition of capital across the capitalist mode of production leads to a tendency for a fall in the rate of profit and crisis. For discussion of the many challenges involved in reviewing such changes, see Kliman (2007). Bhaskar's actual domain is that which is observable and a pertinent example is how a confluence of policies manifest in neoliberal governments' accounting for social housing by its exchange value produced an eviction trap. The observed reality constitutes Bhaskar's empirical domain and the relevant example here is the credit union, its policies and accounting practices in support of homeless prevention. Bhaskar (1998) also distinguishes between transitive and intransitive objects of knowledge. Intransitive objects of knowledge have causes that exist independently of how humans perceive them. In this context, following Marx and Bryer, the intransitive objects of knowledge are labour value categories and the use value of housing. By contrast, transitive phenomena are systems generated by humans to interpret events which, in this context, include Bryer's phenomenal forms of monetary categories in capitalist accounts that manifest in the rent arrears of those caught in an eviction trap.

A case study constitutes research of a phenomenon, event, policy or organization in its own right, rather than as representative of a wider population (Stake, 2005). Case study research often involves using a range of techniques to gather relevant evidence (e.g., Marginson, 2008). There are two broad approaches to case study research. Experimental designs inform the first and require the whole research to be defined in advance and executed accordingly to allow the individual case to contribute to theoretical generalizations (Yin, 2018). Ethnography informs the second approach which accepts that exact advanced design is impossible and research questions are refined and new relevant sources of evidence emerge during the research to allow provision of a particularized explanation of the phenomenon studied (Lee & Saunders, 2017). This research adopted the second approach. Between December 2017 and December 2021, the first author collected and collated evidence from different sources. These included interviews with key managers at Selond, informal conversations with others, documents, continuous email and telephone contact with the second author, details on the UK's BBC series A Matter of Life and Debt, observations through site visits to Selond's offices and participation in some of Selond's events. Relevant documents from or about local councils and social housing providers were also collected. The second author is a manager at Selond. The two authors worked together at the analysis stage to synthesise evidence with academic works to cocreate this consideration of the credit union's homeless prevention initiatives to counter the eviction trap. This evidence is reported below as (i) background details of Selond and its homeless prevention interventions (ii) consideration of Selond's accounting calculations to assess applications for assistance and the range of help provided to help tenants retain the use value of their homes, (iii) Selond's loans to help some tenants honour exchange value obligations pertaining to their homes and (iv) the accountability mechanisms and accounting controls employed after loans were made, to protect Selond's ongoing capability to help tenants threatened by eviction.

B. Lee and L. Carlisle

4.1. The history of Selond and the origins of its housing interventions

Credit unions feature less prominently in Britain than in other advanced economies, due, partly, to their late recognition more than a century after high street banks were established. In contrast to Australia, Canada and the USA where 30.65%, 42.58% and 60.96% of the respective financially registered population held credit union accounts in 2021, the percentage was only 3.26% in the UK (WOCCU, 2022). British credit unions may be seen as originating as a social movement against financial exclusion as they often drew their initial members from people whose poverty prevented them obtaining commercial bank accounts. Consequently, many credit unions grew in poorer areas of cities (Jones, 2008). That is the case with *Selond*. While London residents have the highest average household income in the UK (Office for National Statistics, 2021), "London is one of the most unequal cities on the planet" (Watt, 2021, p. 2). High income is concentrated in the West of the city. *Selond* is in the South-East of London, embracing boroughs with the highest deciles of multiple deprivation (Jones & Ellison, 2011). The *1979 Credit Union Act* (HMSO, 1979) that gave credit unions their own legal status, imposed many restrictions, limiting membership size to 5,000 individuals and excluding organizational and corporate membership. Some restrictions have since been relaxed allowing credit unions to grow through merger. *Selond* was formed in 1992 "by the 20 Christian churches in South-East London" (Manager C). Later, it merged with other local credit unions and developed links with forty corporate organizations including seven registered social housing providers whose employees and tenants may join *Selond*. By 2021, *Selond* operated from four sites and had 12,500 adult members and 1,750 junior members belonging to "savings clubs ... in fifteen schools" (Manager A). *Selond* held f10 million in assets with just over half of those monies out on loan in 2021.

The 1979 Credit Union Act (HMSO, 1979, Section 1.3) installed an enduring requirement for credit unions to promote thrift. Credit unions embodied this provision in practices of expecting members to save before applying for loans. This provision excluded some poor members of a community, historically. Financial difficulties during the neoliberal period led many poor people to borrow from subprime lenders charging high levels of interest (Clifton, 2013). The Government provided a Credit Union Growth Fund (CUGF) in 2005 to allow some such borrowers to migrate to credit unions (Evans & McAteer, 2013; Lee & Brierley, 2017). Participating credit unions received finance to offer loans to new members who had not saved previously. The CUGF project also led to an Eligible Loans Deduction Scheme whereby credit unions could apply to the Department for Work and Pensions for repayment of loans direct from benefits if borrowers defaulted. *Selond* participated in the CUGF project and developed ways of managing accounts of members who had not saved previously. These included introduction of a *Save As You Borrow* loan account that entailed new members' regular payments both repaying the initial loan and contributing to savings that could not be redeemed until the loan was repaid. After the CUGF scheme ended, many people in *Selond*'s constituency continued to suffer poor living standards. For example, around 24% or 13,000 people in one local borough earned below the living wage in 2019, so many tenants struggled with rent arrears following the benefit changes reported in section 3.

Threats of eviction were problematic for local councils as they retained a statutory obligation to provide advice and temporary accommodation to homeless local people, or those at risk of homelessness. In the context of *Selond*'s expertise in handling loan accounts of people without savings, one council provided £85,000 in 2010 for *Selond* to offer a homeless prevention loan (HPL). As Bryer (2017) has pointed out, provision of products and services requires productive labour to create use value, but it will also often involve unproductive labour. Much of the use value of housing is created at the time of construction and added to, or reconstituted, periodically by home improvements. The local council employed considerable amounts of unproductive labour to find people new residence after avoidable evictions. Manager A reported "if somebody loses their home, evicted, for example, the total cost to the council is about £28,000 to rehouse that family", whereas an internal document of the local council's housing prevention team reported that each HPL intervention – for which *Selond* received a small administrative fee – cost only around 3% of the estimated £28,000 of each eviction, thus, reducing the amount of unproductive labour at the local council. That scheme's success prompted the second local council to support introduction of a similar scheme two years later. Provision of funds for HPLs led to *Selond*'s other homeless prevention interventions.

4.2. Overview of types of assistance provided by Selond for tenants threated with eviction

Selond's initiatives to support tenants may be interpreted as matching different ways of protecting a tenant's utilisation of the use value of his/her home, to the scale of his/her means to pay the rental exchange values of accommodation. *Selond*'s homeless prevention interventions sometimes started with a landlord of a tenant in arrears applying to a court for an eviction order to be issued. Many tenants under threat of eviction approached their local council which referred them to *Selond*. Manager B explained that tenants under the threat of eviction were not the only ones with rent arrears but instead were "the tip of the iceberg. They are the people that are getting evicted next week. They've got their eviction letter, the bailiffs are coming around". As registered financial institutions, credit unions are obliged to conduct affordability checks before providing loans. Each applicant for a loan was, thus, asked to complete a form documenting his/her income and expenditure to establish whether s/he had sufficient financial resources for repayments after meeting other obligations. The form was scrutinised by a *Selond* employee before an interview took place to ensure the form's contents were accurate. Manager A explained: "[W]e've got four underwriters and about 15 staff who all know what to look for, what questions to ask [when assessing loan applications]. So we do prod them [i.e., prospective borrowers] deeply." Applicants were only eligible for a loan if they had a disposable income of £240 per month after existing necessary expenditure. This highlights the tension between accounting for loans according to the tenant's capacity to meet the terms of the exchange and the tenant's need for the use value of their home. Consequently, few applicants received loans. For example, in 2019, in one borough, only 35 out of 209 applicants received a HPL. *Selond* developed other means to help others.

Out of the 209 applicants, 100 who did not face an immediate threat of eviction were given informal advice to improve their

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

capability to retain the use value of their homes. Manager B explained how *Selond* sought to identify the cause of any financial difficulties that a prospective applicant might be experiencing and: "Try and signpost them to organisations that might be able to help them. [We ask:] "Are you getting the right benefits? Do you know what benefits you're entitled to? Did you know that you can go to these organisations?"" Another 74 who failed *Selond*'s affordability criteria for a HPL but faced a greater threat of eviction and loss of the use value of their homes were advised to negotiate with their landlords. In some instances, *Selond* provided guidance around the unproductive labour and costs that the landlord might incur if the application for a Court Order was contested. *Selond* also advised tenants about (i) the court application form, N244, which would lead to a contested hearing over the eviction and (ii) the best way to present a case contesting eviction. Manager B explained how those who did not have sufficient disposable income to receive a HPL might be advised to write to the landlord offering to pay a small, affordable sum such as £25 monthly in addition to rent, to reduce the arrears which would help the tenant to avoid eviction. He said: "I help them to write the letter. They write to their landlord. If the landlord rejects that and they still want to evict them but their rent arrears are coming down, because you get a rent statement in front of the judge, that housing association will lose the minute he's in front of the judge." In such instances, landlords were often amenable to negotiating with a tenant over settling for a limited sum of the arrears if the tenant committed to maintaining rent payments in the future. The remaining 35 applicants were eligible for the HPLs which are now discussed.

4.3. Selond's homeless prevention loans to help tenants honour exchange value obligations

The first HPL scheme was inaugurated in March 2010. The grant from the local council was for *Selond* to provide interest free loans to enable recipients to repay rent arrears and retain use value of their homes. As credit unions may only lend money to members, loan applicants had to join *Selond*. Members' loans were paid directly to the landlord – often by instalments with this first scheme – so that rent arrears were reduced incrementally and cleared eventually. The principle of saving embedded in *Selond*'s other financial services for members without prior savings were applied to the HPLs. Thus, some new member's agreed periodic payments contributed to both repayment of the loan and a savings component and a number continued to save after the HPL was repaid. Table 2 below shows the number of loans from the scheme's launch in 2010 until the end of 2021. The increase in applications in the second half of the period was a legacy of reductions in benefits, discussed in section 3.

Expertise acquired through participation in the CUGF Project, of building safeguards around repayment into the original loan agreement, allowed *Selond* to manage the HPLs. These safeguards are shown in Fig. 1, below, which is an extract from *Selond*'s standard loan agreement. *Selond*'s knowledge that a loan was used to protect a borrower's use value of an address that *Selond* knew, supplemented the security provided by the loan agreement.

Significantly, from the inception of the scheme in 2010 until December 2021, 189 families received loans from the HPL scheme, helping them to retain the use value of their homes. Repayment of loans by most borrowers provided funds for provision of more loans under the HPL scheme to others at risk of eviction. Consequently, the £366,784 total of loans was more than four times the original £85,000 grant provided to establish the scheme and the £40,521 reported as written off is of considerably lesser value than the original £85,000 grant.

Table 3, below, shows the details of the subsequent scheme that *Selond* administered for a neighbouring borough. The scheme enabled 246 families to continue to enjoy the use value of their homes. Clearly, the 29.7% of loans shown as written off in the second scheme is more than double the 14% of all loans shown as written off in the first scheme. While much of this is attributable to the first three years of the scheme when there were problems with 22 of 52 or over 42% of the loans provided, the 51 of 194 or over 26% of loans in the seven subsequent years is higher than the percentage of defaults in the first scheme. These variations are attributable, in part to the whole loan in the second scheme being paid to the landlord at the outset, providing less incentive for the tenant to engage with *Selond* to repay the debt. Manager B said that the first scheme "was similar to teaching tenants to fish by paying off the arrears in instalments while they learn to budget by saving and repaying a loan" whereas the second scheme "was like giving the tenant a fish by paying off all the arrears at the start of the loan and they do not learn to budget". The next subsection addresses recovery of debts.

Year	Number of loans	Amount on Loan	Repaid to date	Outstanding balance to date	Value of actual write-off	Number of write-offs
2010	9	30,370	29,950	0	420	1
2011	12	30,840	13,375	0	17,465	8
2012	4	13,165	13,165	0	0	0
2013	8	21,330	17,127	0	4,203	1
2014	10	14,870	10,370	2,000	2,500	2
2015	15	37,520	34,863	0	2,657	3
2016	29	40,670	37,054	0	3,616	4
2017	18	32,895	29,430	0	3,465	3
2018	26	53,135	48,285	205	4,645	3
2019	35	63,182	53,952	9,230	0	0
2020*	15	19,637	12,952	5,135	1,550	2
2021*	8	9,170	2,151	7,019	0	1 loan at risk
Totals	189	366,784	302,674	23,589	40,521	27
%	100	100	82.5	6.4	11	14

Selond's Homeless prevention loans 2010 to 2021 in the first borough.

Table 2

* Figures in 2020 and 2021 affected by lockdowns and provisions of Coronavirus Act (2020).

- 6. In the event at anytime Loan repayments fall more than three months in arrears the Loan shall be in default. The whole of the Loan plus interest will become immediately repayable and the Lender will be entitled to apply the value of the Borrower's shares to the outstanding Loan balance and any unpaid interest.
- 7. The Borrower will be liable for any costs the Lender incurs in recovering sums due under this Agreement, such as the fees of any debt recovery agency and court costs.
- 8. The Borrower agrees that if the Loan is in default, information about the Loan may be passed on to the Department for Work & Pensions for their consideration of deductions that the Borrower is or will be entitled to.
- The borrower agrees that the information about the Loan repayments and any defaults of the repayments for this loan may be shared with the Local Government authorities. Where appropriate this will be shared with their registered agents.

Fig. 1. Extract from loan agreement protecting Selond's provision of loan.

 Selond's homeless prevention loans 2012 to 2021 in the second borough.

Year Number of loans		Amount on Loan	Repaid to date	Outstanding balance to date	Value of actual write-off	Number of write-offs	
2012	5	6,250	3,370	0	2,880	2	
2013	31	29,162	17,562	0	11,600	15	
2014	16	11,549	8,925	0	2,624	5	
2015	25	42,972	37,884	0	5,088	7	
2016	26	39,765	33,678	0	6,087	7	
2017	43	77,201	62,915	3,364	10,922	9	
2018	29	79,248	55,287	0	23,961	11	
2019	54	160,258	76,271	56,476	27,511	13	
2020*	12	33,445	12,588	14,965	5,892	3	
2021*	5	7,150	1,837	5,313		1 loan at risk	
Totals	246	487,000	310,317	80,118	96,565	73	
%	100	100	63.7	16.5	19.8	29.7	

* Figures in 2020-1 affected by lockdowns and provisions of Coronavirus Act (2020).

4.4. Accountability mechanisms to protect Selond's ongoing capability to help

Repayment of the loans allowed Selond to help others suffering the threat of eviction. It was, thus, important that Selond recovered the funds from the loans provided. Selond had different types of relationship with others involved in the respective schemes, which affected its ability to recover loans. Manager B described the first scheme as "a true partnership between us [i.e., Selond], the tenant, the social housing provider and the local council". This made it relatively easy for Selond to acquire information about difficulties encountered by the borrower. By contrast, Manager B said that in the second scheme "it is very difficult to monitor what's happening with the tenants because others don't share information". The different type of relationship may arise in part from the different constitution of the social landlords involved in the respective schemes. After the 1986 Housing and Planning Act allowed councils to transfer their housing to others, different types of arrangements followed including transfer of ownership, or creation of Arms-Length Management Organizations (ALMOs) whereby a not-for-profit third-party manages the properties on behalf of a council (see Watt, 2021, pp 56-8). In the first scheme, the local council retained ownership of the properties that were managed by an ALMO. By contrast, the other local council transferred ownership of the residences to the second social housing provider. The second social housing provider remains a not-for-profit organization, but now belongs to a wider group that includes some commercial companies. The group has both investment organizations and a development wing involved in commercial building projects to construct houses for market sale. Consistent with the advent of financialization as a recent development in the neoliberal era (Smyth et al., 2020), the social housing landlord associated with the second scheme issued bonds to realise some of its objectives. Notably, this organization was now more prone to pursue evictions, although - as noted above - Selond gave advice to tenants threatened with eviction.

Regardless of other factors, *Selond* sought repayment of loans. In accord with conventions regulating what Bryer describes as the phenomenal forms of financial accounts in capitalist economies, *Selond* made provisions for bad debts and wrote them off formally on its balance sheet. However, *Selond* did not exclude the possibility of recovering such monies. *Selond* used its accounting system to distinguish between borrowers who did not have the means to repay loans and those who might have the means to repay loans but were choosing not to. One way in which this distinction was realized was by identifying high value loans provided to tenants – decided originally by affordability criteria – and the number and value of repayments made before repayment stopped. In some such instances – with the preponderance being borrowers under the second scheme where the entire loan was provided at the outset – the only repayment received was when the borrower joined *Selond* to acquire the loan. *Selond* categorised the debtors into the deceased from whom money could not be collected, borrowers protected by "debt relief orders" where legal mechanisms prevented collection of the

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

debt and others. Those others were pursued using the Eligible Loans Deduction Scheme. 10% of the written-off debts had already been collected from this group and monies were continuing to be collected. The hybrid position of *Selond* meant that while it was duty-bound to remain a viable financial institution and recover monetary debts, it used those funds to provide HPLs to help others protect the use value of their homes.

5. Concluding discussion

This discussion started by recognizing the dualistic quality of domestic residences in capitalist economies, both as homes with use values as safe havens for people and as commodities in markets (Marx, 1954; Wilde, 2020). This dualism creates the possibility of people being evicted from their homes if they do not meet the financial commitments associated with such a commodity. The first contribution of this article has been to apply Gramsci's (1978) concept of the historic bloc in the context of accounting. A historic bloc refers to an alliance of forces around a dominant class that helps in the articulation across the State, civil society and the economy, of the conditions for the development of the economy under the direction of the dominant class according to the relative strength of political forces. The concept of historic bloc has been used to understand changes in the State's emphasis when accounting for social housing. There have been markedly different approaches to accounting for social housing between the time of a social democratic historic bloc between the 1940 s and 1970 s when ordinary workers had greater political support and the emphasis was on need and use values, and the period of the neoliberal historic bloc since the late 1970 s when political support for workers was less and the emphasis shifted to exchange values.

This discussion's second contribution is to synthesise works of different Marxist writers (Bryer, 2017, 2019; Cooper et al., 2005; Lee & Cassell, 2008) on social accounts. Lee and Cassell suggest social accounts may be used to transform civil society institutions to be prefigurative of an alternative society in preparation of the formation of a new historic bloc. Cooper et al.'s (2005) advocacy of social accounts suggests a strategy for pursuing such a transformation. This will entail preparing social accounts away from the influence of markets and management, using critical theory to locate violations of rights within the contradictory relationships of a capitalist economy that caused the violation and linking that violation to the struggles of social movements. This article highlights how the organization of capitalist economies leads to evictions that are violations of people's rights to a safe haven. Highlighting the causes of such violations has been helped by Bryer's work that proposes the critical theory employed should include Marx's economic writings. In particular, Bryer suggests that social accounts should highlight use values and how the monetary forms of categories such as "profits", "costs", "dividends", etc., associated with capitalist relations of production, provide fetters on the production of use values.

A third contribution of this article is the development of the concept of the eviction trap. Particular attention has been paid to how governments' accounting for social housing by its use value during the social democratic period was marked by subsidisation of a comprehensive building programme to meet citizens' needs, legal constraints on the sale of council houses and regulation of rents to ensure residential properties were affordable for their occupants. In contrast, governments in the neoliberal period accounted for social housing by its exchange value. This has manifest in policies of considerably reduced support for construction of council houses with use values and raising rents towards market rates which, when coupled with reduced incomes, has placed less affluent tenants in jeopardy of eviction because of an inability to pay the rental exchange values of their residences, particularly in places of premium-priced property such as London. Consistent with Bryer's ideas, it has been demonstrated how governments' shifting focus in accounting for housing from use values to exchange value provides fetters on construction of houses despite there being considerable need.

The final contribution of this discussion has been an empirical one of reporting and analysing the different initiatives taken by a credit union to ameliorate the pressures of the eviction trap for some social housing providers' tenants unable to meet the rental exchange values of their properties. While credit unions are hybrid organizations situated across both the economy and civil society, their place in civil society constitute credit unions as a social movement comprising membership cooperatives serving members of a specific community, many of whom suffered financial exclusion historically. The research showed that in the case study of Selond, this role was extended to people threatened with housing exclusion. Thus, consistent with Cooper et al.'s recommendations, the account provided here has been linked to the struggles of a social movement, although Selond's intersectional position and regulation as an financial institution meant that the assistance provided was sometimes limited to provision of advice. Nevertheless, in many instances, this advice was instrumental in preventing evictions. Selond offered loans to others who satisfied assessments that the loans are affordable. These loans are underwritten by local councils and helped to protect tenants' enjoyment of their homes' use value. The number of people in need of such help is likely to increase with the 2022-3 "cost of living crisis". Some may argue that Selond's initiatives had minimal effect. They have saved only 435 families from the trap of eviction over a decade and provided advice to others who had rent arrears in just two London boroughs. Such assistance does not address the needs of those without social housing but require it. The number receiving help may be dwarfed by the needs of the 0.5% of several hundred thousand social housing tenants losing the use value of their homes through eviction every year (Clark et al., 2017). However, the impacts of Selond's homeless prevention initiatives for the families involved are considerable. Their impact would be multiplied if all credit unions with links to housing bodies copy these initiatives. Of course, such actions would be insufficient to redress the damage to public housing provision during the neoliberal period.

The 1980 Housing Act marked advent of the neoliberal historic bloc's housing policies and illustrated the hallmark of hegemony by winning consent for a privatized exploitative economy from those who bought council houses at discounted prices while increasing rents to contribute to the threat of eviction for others. The eviction trap will only be eliminated by reversing the increasing disparities in incomes, effective regulation to limit rent increases and a sustained programme of social housing construction at rents that tenants can afford. This will help ensure social housing's use value is maximised, but it will require composition of a new historic bloc which will not be an easy task. Civil society may be becoming increasingly complex as manifest in credit unions that were formed as

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

alternative types of membership co-operative to counter the financial exclusion created by large commercial banks that circulate money to help realisation of surplus value, finding their activities shaped by the same regulations affecting those banks. Building alliances with social movements as vehicles for progressive change to help transform their civil society institutions into bodies prefigurative of a socialist economy will require identifying – and being sensitive to – the conflicting pressures that they face. Execution of the practical changes needed to transition to a socialist economy that prioritizes use values is likely to be arduous and require patience. Construction of houses with use values to satisfy existing needs may take some years and part of the process of change will entail identifying, reducing and progressively eliminating the unproductive labour consumed currently in managing existing, unnecessary scarcities. Untangling the conditions for potential alliances across civil society and identifying positive reductions in unnecessary, unproductive labour – such as the way that a credit union reduced unnecessary pressures on a council's homeless prevention department – are important contributions to social accounts to help form the alliances required to construct an alternative historic bloc. Similarly, managing housing shortages while residences are created for use value to find solutions that are more humane than penalising a resident for having a spare bedroom could be an important contribution to a social account.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The data that has been used is confidential.

Acknowledgements

Bill Lee's collection of empirical work in this article was assisted by BA/Leverhulme Small Research Grants SG163441. Bill Lee acknowledges that funding with thanks. The authors would like to thank *Selond* Credit Union for their participation in this research. The authors would also like to thank the reviewers, participants in the preparatory workshop and the editors of the special issue for their helpful feedback on earlier drafts of this manuscript.

References

Alawattage, C., & Wickramasinghe, D. (2008). Appearance of accounting in a political hegemony. Critical Perspectives on Accounting, 19, 293–339.

- Baker, A. (2020). Eviction as infrastructure. City, 24, 143–150. https://doi.org/10.1080/13604813.2020.1739417
- Bebbington, J., Brown, J., Frame, B., & Thomson, I. (2007). Theorizing engagement: The potential of a critical dialogic approach. Accounting, Auditing & Accountability Journal, 20, 356–381.
- Bhaskar, R. (1975). A Realist Theory of Science [RTS]. Leeds: Leeds Books.
- Bhaskar, R. (1998). Philosophy and Scientific Realism. In M. Archer, R. Bhaskar, A. Collier, T. Lawson, & A. Norrie (Eds.), Critical Realism: Essential Readings (pp. 16–47). London: Routledge.
- Boyne, G.A. and Walker, R.M. (1999). Social Housing Reforms in England and Wales: A Public Choice Evaluation. Urban Studies, 36(13), 2237-2262.
- Brown, J. (2009). Democracy, sustainability and dialogic accounting technologies: Taking pluralism seriously. *Critical Perspectives on Accounting*, 20, 313–342. Bryer, R. A. (2017). Accounting for Value in Marx's Capital: The Invisible Hand. Lanham, MD: Lexington Books.
- Bryer, R. A. (2019). Accounting for History in Marx's Capital: The Missing Link. Lanham, MD: Lexington Books.
- Burawoy, M. (2000). Marxism after communism. Theory and Society, 29, 151-174.

Carchedi, G., & Roberts, M. (2013). The long roots of the present crisis: Keynesians, Austerians, and Marx's law. World Review of Political Economy, 4(1), 86–115.
Carroll, A. B., & Shabana, K. M. (2010). The business case for corporate social responsibility: A review of concepts, research and practice. International Journal of Management Reviews, 12, 85–105.

- Catchpowle, L., & Smyth, S. (2016). Accounting and social movements: An exploration of critical accounting praxis. Accounting Forum, 40, 220–234.
- Clark, A., Hamilton, C., Jones, M., & Muir, K. (2017). Poverty, evictions and forced moves. York: Joseph Rowntree Foundation.

Clifton, H. (2013). Drowning in debt: How irresponsible lenders are creating a tidal wave of misery. Salford: Church Action on Poverty.

- Cohen, J. L. (1983). Class and civil society: The limits of Marxian critical theory. Oxford: Martin Robinson.
- Cohen, J. L., & Arato, A. (1992). Civil society and political theory. Cambridge, Massachusetts: MIT Press.

Cooper, C., Taylor, P., Smith, N., & Catchpowle, L. (2005). A discussion of the political potential of Social Accounting. *Critical Perspectives on Accounting*, *16*, 951–974. Cooper, C. (1995). Ideology, hegemony and accounting discourse: A case study of the National Union of Journalists. *Critical Perspectives on Accounting*, *6*, 175–209. Cooper, C., & Johnston, J. (2012). Vulgate accountability: Insights from the field of football. *Accounting*, *Auditing & Accountability Journal*, *25*, 602–634. Crisis (2018). *EVERYBODY IN: How to end homelessness in Great Britain*. London: Crisis.

Daly, G., & Lund, B. (2020). Housing Policy. In H. Bochel, & G. Daly (Eds.), Social Policy (pp. 405-436). Abingdon, Oxon: Taylor and Francis Group.

Department for Community and Local Government. (2014). Guidance on Rents for Social Housing. London: Department for Community and Local Government. Dey, C., Russell, S., & Thomson, I. (2011). Exploring the potential of shadow accounts in problematising institutional conduct. In A. Ball, & S. P. Osborne (Eds.), Social accounting and public management: Accountability for the common good (pp. 64–75). Abingdon, Oxon: Routledge.

Ejiogu, A., Ambituuni, A., & Ejiogu, C. (2021). Accounting for accounting's role in the neoliberalization processes of social housing in England: A Bourdieusian perspective. *Critical Perspectives on Accounting*, *80*, Article 102053.

Ejiogu, A., & Denedo, M. (2021). Stigma and Social Housing in England. Durham: Durham University Business School.

Elkington, J. (1997). Cannibals with forks: The triple bottom line of 21st century business. London: Capstone.

Evans, G., & McAteer, M. (2013). Can payday loan alternatives be affordable and viable? An evaluation of London Mutual Credit Union's pilot scheme. London: The Financial Inclusion Centre.

Fée, D. (2009). Acknowledging the limits of Thatcherism: Housing policies during the Major years. Observatoire de la société britannique, 7, 233-248.

Garrett, H., Nicol, S., & Mackay, M. (2019). 100 Years of Council Housing. London: BRE Trust.

Goddard, A. (2002). Development of the accounting profession and practices in the public sector—a hegemonic analysis. Accounting, Auditing & Accountability Journal, 15, 655–688.

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

Goddard, A. (2005). Reform as regulation—accounting, governance and accountability in UK local government. Journal of Accounting & Organisational Change, 1, 27-44.

Gov.UK (2022). Live tables on housing supply: indicators of new supply. Retrieved from https://www.gov.uk/government/statistical-data-sets/live-tables-on-house-building. Accessed September 28, 2022.

- Gramsci, A. (1978). Selections from the Prison Notebooks. London: Lawrence & Wishart.
- Gray, R. (2002). The social accounting project and accounting organizations and society: Privileging engagement, imaginings, new accountings and pragmatism over critique? Accounting, Organizations and Society, 27, 687–707.

Gray, R., Bebbington, J., & Collison, D. (2006). NGOs, civil society and accountability: Making the people accountable to capital. Accounting, Auditing & Accountability Journal, 19, 319–348.

Gray, R., Brennan, A., & Malpas, J. (2014). New accounts: Towards a reframing of social accounting. Accounting Forum, 38, 319-348.

Gray, R., Owen, D., & Adams, C. (1996). Accounting and accountability: Changes and challenges in corporate social and environmental reporting. Harlow: Prentice Hall. Haines-Doran, T. (2022). Critical accounting scholarship and social movements: The case of rail privatisation in Britain. Critical Perspectives on Accounting. ISSN 1045-2354.

Hall, S. (1987). The Great Moving Right Show. In S. Hall (Ed.), The Hard Road to Renewal: Thatcherism and the Crisis of the Left (pp. 39-56). London: Verso.

Harris, J. (2016). The end of council housing. *The Guardian*, Monday 4th January, 2016. Retrieved from https://www.theguardian.com/society/2016/jan/04/end-ofcouncil-housing-bill-secure-tenancies-pay-to-stay. Accessed January 30, 2022.

Harvey, D. (2005). A Brief History of Neoliberalism. Oxford: Oxford University Press.

HMG. (2001). The Financial Services and Markets Act 2000 (Carrying on Regulated Activities by Way of Business) Order 2001. London: National Archives.

HMSO. (1979). The Credit Union Act. London: HMSO.

Jones, P. A. (2005). Philanthropy and Enterprise in the British Credit Union Movement. Economic Affairs, 25(2), 13–19.

Jones, P. A. (2008). From tackling poverty to achieving financial inclusion: The changing role of British credit unions in low income communities. *The Journal of* SocioEconomics, 37, 2141–2154.

Jones, P. A., & Ellison, A. (2011). Community finance for London: Scaling up the credit union and social finance sector. Liverpool: Faculty of Health and Applied Social Sciences, LJMU.

Kliman, A. (2007). Reclaiming Marx's Capital: A Refutation of the Myth of Inconsistency. Plymouth: Lexington Books.

Kliman, A. (2012). The Failure of Capitalist Production: Underlying Causes of the Great Recession. London: Pluto Press.

Kliman, A. (2015). The Great Recession and Marx's crisis theory. American Journal of Economics and Sociology, 74, 236-277.

Kliman, A. (2017). Tendency of the Rate of Profit to Fall: Long-term Dynamics. In D. M. Brennan, D. Kristjanson-Gural, C. P. Mulder, & E. K. Olsen (Eds.), Routledge Handbook of Marxian Economics (pp. 225–233). London: Routledge.

Kliman, A., & Williams, S. D. (2015). Why 'financialisation' hasn't depressed US productive investment. Cambridge Journal of Economics, 39, 67–92.

Labour Party (1945). 1945 Labour Party Election Manifesto: Let Us Face the Future: A Declaration of Labour Policy for the Consideration of the Nation. Retrieved from http://www.labour-party.org.uk/manifestos/1945/1945-labour-manifesto.shtml. Accessed November 8, 2021.

Laffin, M. (2019). Explaining reforms: Post-New Public Management myths or political realities? Social housing delivery in England and France. International Review of Administrative Sciences, 85(1), 45–61.

Lapsley, I. (2008). The NPM agenda: Back to the future. Financial Accountability and Management, 24, 77-96.

Lee, B. (2010). The individual learning account experiment in the UK: A conjunctural crisis? Critical Perspectives on Accounting, 21, 18-30.

Lee, B., & Brierley, J. (2017). UK government policy, credit unions and payday loans. International Journal of Public Administration, 40, 348–360.

Lee, B., & Cassell, C. (2008). Employee and social reporting as a war of position and the union learning representative initiative in the UK. Accounting Forum, 32, 276–287.

Lee, B., & Cassell, C. (2017). Facilitative reforms, democratic accountability, social accounting and learning representative initiatives. Critical Perspectives on Accounting, 46, 24–37.

Lee, B., & Saunders, M. N. K. (2017). Conducting Case Studies. London: Sage.

Lees, L., & White, H. (2020). The social cleansing of London council estates: Everyday experiences of 'accumulative dispossession'. Housing Studies, 35, 1701–1722. https://doi.org/10.1080/02673037.2019.1680814

Lehman, C. (1995). Accounting's changing role in social conflict. London: Paul Chapman Publishing.

Lehman, C., & Tinker, T. (1987). The 'real' cultural significance of accounts. Accounting, Organizations and Society, 12, 503-522.

Lehman, G. (1999). Disclosing new worlds: A role for social and environmental accounting and auditing. Accounting, Organizations and Society, 24, 217-241.

Leversha, G. (1977). Beyond spontaneity. In J. Bloomfield (Ed.), Class, Hegemony and Party (pp. 109-125). London: Lawrence & Wishart.

Marginson, D.E.W. (2008). The Case Study, The Interview and The Issues: A Personal Reflection. In Humphrey, C. and Lee, B. (Eds.), *The Real Life Guide to Accounting Research* (pp. 325-337), paperback edition. Oxford: Elsevier in conjunction with CIMA publishing.

Marx, K. (1954). Capital (Volume 1). London: Lawrence & Wishart.

Marx, K. (1959). Capital (Volume 3). London: Lawrence & Wishart.

Modell, S. (2009). In defence of triangulation: A critical realist approach to mixed methods research in management accounting. Management Accounting Research, 20, 208–221.

Modell, S. (2017). Critical realist accounting research: In search of its emancipatory potential. Critical Perspectives on Accounting, 42, 20-35.

Murphy, L. R. (1970). Rebuilding Britain: The Government's Role in Housing and Town Planning. 1945-57. The Historian, 32(3), 410-427.

NAO (2021). Regulation of private renting. REPORT [Summary] by the Comptroller and Auditor General. SESSION 2021-22, 10 DECEMBER 2021, HC 863. Retrieved from https://www.nao.org.uk/wp-content/uploads/2021/09/Regulation-of-private-renting-Summary.pdf. Accessed December 12, 2021.

NHF (2020). People in Housing Need: A comprehensive analysis of the scale and shape of housing need in England today, 15 September 2020. London: National Housing Federation. Retrieved from https://www.housing.org.uk/globalassets/files/report-people-in-housing-need-final.pdf. Accessed January 30, 2022.

OECD (2021). OECD Affordable Housing Database. Retrieved from https://www.oecd.org/housing/data/affordable-housing-database/. Accessed January 26, 2022. Office for National Statistics (2021). What are the regional differences in income and productivity? Retrieved from https://www.ons.gov.uk/visualisations/dvc1370/. Accessed May 24, 2021.

Regulator of Social Housing (2021). Guidance Value for Money Standard - April 2018, Updated 3 June 2021. Retrieved from https://www.gov.uk/government/ publications/value-for-money-standard/value-for-money-standard-april-2018. Accessed January 26, 2022.

Richardson, A. (1989). Corporatism and intra-professional hegemony: A study of regulation and internal social order. Accounting, Organizations and Society, 14, 415–431.

Roberts, M. (2016). The Long Depression: How It Happened and What Happens Next. Chicago: Haymarket Books.

Sikka, P., Wearing, D., & Nayek, A. (1999). No accounting for exploitation. Essex: Association for Accounting & Business Affairs.

Simon, R. (1982). Gramsci's political thought. London: Lawrence & Wishart.

Smyth, S. (2012a). Contesting public accountability: A dialogical exploration of accountability and social housing. Critical Perspectives on Accounting, 23, 230–243.

Smyth, S. (2012b). The privatization of council housing: Stock transfer and the struggle for accountable housing. Critical Social Policy, 33, 37–56.

Smyth, S., Cole, I., & Fields, D. (2020). From gatekeepers to gateway constructors: Credit rating agencies and the financialisation of housing associations. *Critical Perspectives on Accounting*, 71, Article 102093.

Smyth, S., & Whitfield, D. (2017). Maintaining market principles: Government auditors, PPP equity sales and hegemony. Accounting Forum, 41, 44–56.

Spence, C. (2009). Social accounting's emancipatory potential: A Gramscian critique. Critical Perspectives on Accounting, 20, 205-227.

Stake, R. E. (2005). Qualitative case studies. In N. K. Denzin, & Y. S. Lincoln (Eds.), The Sage handbook of qualitative research (third edition, pp. 443–446). Thousand Oaks. CA: Sage.

Thomson, I., Dey, C., & Russell, S. (2015). Activism, arenas and accounts in conflicts over tobacco control. Accounting, Auditing & Accountability Journal, 28, 809–845.

CLE DD

B. Lee and L. Carlisle

Critical Perspectives on Accounting xxx (xxxx) xxx

UK Housing Review (2021). COMPENDIUM. Retrieved from https://www.ukhousingreview.org.uk/ukhr21/compendium.html. Accessed March 25, 2022. Watt, P. (2021). Estate Regeneration and Its Discontents: Public Housing, Place and Inequality in London. Bristol: Policy Press. Wilde, M. (2020). Eviction, Gatekeeping and Militant Care: Moral Economies of Housing in Austerity London. Ethnos. DOI: 10.1080/00141844.2019.1687540. WOCCU (undated). The Credit Union Difference. Retrieved from https://www.woccu.org/impact/credit_unions. Accessed July 28, 2021. Woccu (2022). 2021 Statistical Report. Washington, USA: World Council of Credit Unions.
 Yin, R. K. (2018). Case Study Research and Applications Design and Methods (6th Edition). Los Angeles, CA: Sage.
 Zamfirescu, I. and Chelcea, L. (2020). Evictions as infrastructural events. Urban Geography. DOI: 10.1080/02723638.2020.1778281.